Author's response to reviews

Title: Automated inter-rater reliability assessment and electronic data collection in a multi-center breast cancer study

Authors:

Soe Soe Thwin (sst@bu.edu)
Kerri M Clough-Gorr (kmclough@bu.edu)
Maribet C McCarty (Maribet.C.McCarty@HealthPartners.Com)
Timothy L Lash (tlash@bu.edu)
Sharon H Alford (SALFORD1@hfhs.org)
Diana S.M. Buist (buist.d@ghc.org)
Shelley M Enger (Shelley.M.Enger@kp.org)
Terry S Field (FIETE01@fallon-clinic.com)
Floyd Frost (FFrost@Irri.org)
FeiFei Wei (Feifei.X.Wei@HealthPartners.Com)
Rebecca A Silliman (rsillima@bu.edu)

Version: 7 Date: 15 June 2007

Author's response to reviews: see over
Dear Editor,

We have uploaded the final revised version of the manuscript titled, “Automated inter-rater reliability assessment and electronic data collection in a multi-center breast cancer study,” for publication in your Medical Research Methodology online journal. We have included the corresponding author’s information in the manuscript, as requested.

Thank you very much for your consideration and your patience.

Sincerely,

Soe Soe Thwin

Corresponding Author:
Soe Soe Thwin
Geriatrics Section, Boston Medical Center
88 E. Newton St.
Boston, MA 02118
(t) 617-638-8986
(f) 617-638-8312
sst@bu.edu